# **UC Irvine**

# **UC Irvine Previously Published Works**

# **Title**

Prospective Observational Study of Pazopanib in Patients with Advanced Renal Cell Carcinoma (PRINCIPAL Study)

# **Permalink**

https://escholarship.org/uc/item/8xg4q8bp

# Journal

The Oncologist, 24(4)

## **ISSN**

1083-7159

# **Authors**

Schmidinger, Manuela Bamias, Aristotelis Procopio, Giuseppe et al.

# **Publication Date**

2019-04-01

# DOI

10.1634/theoncologist.2018-0787

Peer reviewed



# Prospective Observational Study of Pazopanib in Patients with Advanced Renal Cell Carcinoma (PRINCIPAL Study)

Manuela Schmidinger,<sup>a</sup> Aristotelis Bamias,<sup>b</sup> Giuseppe Procopio,<sup>c</sup> Robert Hawkins,<sup>d</sup> Angel Rodriguez Sanchez,<sup>e</sup> Sergio Vázquez,<sup>f</sup>
Narayanan Srihari,<sup>g</sup> Haralabos Kalofonos,<sup>h</sup> Petri Bono,<sup>i</sup> Chaitali Babanrao Pisal,<sup>j</sup> Yulia Hirschberg,<sup>k</sup> Luca Dezzani,<sup>k</sup> Qasim Ahmad,<sup>k</sup>
Eric Jonasch,<sup>I</sup> on behalf of the **PRINCIPAL** Study Group

<sup>a</sup>Medical University of Vienna, Vienna, Austria; <sup>b</sup>National and Kapodistrian University of Athens Alexandra Hospital, Athens, Greece; <sup>c</sup>Fondazione IRCCS Istituto Nazionale dei Tumori, Milan, Italy; <sup>d</sup>The Christie Hospital and University of Manchester, Manchester, United Kingdom; <sup>e</sup>University Hospital of Leon, Leon, Spain; <sup>f</sup>Hospital Universitario Lucus Augusti, Lugo, Spain; <sup>g</sup>Shrewsbury & Telford Hospitals NHS Trust, Shrewsbury, United Kingdom; <sup>h</sup>University of Patras, Patras, Greece; <sup>i</sup>Helsinki University Hospital and University of Helsinki, Finland; <sup>j</sup>Novartis Healthcare Private Limited, Hyderabad, India; <sup>k</sup>Novartis Pharmaceuticals Corporation, East Hanover, New Jersey, USA; <sup>1</sup>The University of Texas MD Anderson Cancer Center, Houston, Texas, USA *Disclosures of potential conflicts of interest may be found at the end of this article.* 

Key Words. Renal cell carcinoma • Pazopanib • Real-world • Observational • Tyrosine kinase inhibitor

#### Abstract.

**Background.** Real-world data are essential to accurately assessing efficacy and toxicity of approved agents in everyday practice. PRINCIPAL, a prospective, observational study, was designed to confirm the real-world safety and efficacy of pazopanib in patients with advanced renal cell carcinoma (RCC).

Subjects, Materials, and Methods. Patients with clear cell advanced/metastatic RCC and a clinical decision to initiate pazopanib treatment within 30 days of enrollment were eligible. Primary objectives included progression-free survival (PFS), overall survival (OS), objective response rate (ORR), relative dose intensity (RDI) and its effect on treatment outcomes, change in health-related quality of life (HRQoL), and safety. We also compared characteristics and outcomes of clinical-trial-eligible (CTE) patients, defined using COMPARZ trial eligibility criteria, with those of non-clinical-trial-eligible

(NCTE) patients. Secondary study objectives were to evaluate clinical efficacy, safety, and RDI in patient subgroups.

Results. Six hundred fifty-seven patients were enrolled and received ≥1 dose of pazopanib. Median PFS and OS were 10.3 months (95% confidence interval [CI], 9.2–12.0) and 29.9 months (95% CI, 24.7 to not reached), respectively, and the ORR was 30.3%. HRQoL showed no or little deterioration over time. Treatment-related serious adverse events (AEs) and AEs of special interest occurred in 64 (9.7%), and 399 (60.7%) patients, respectively. More patients were classified NCTE than CTE (85.2% vs. 14.8%). Efficacy of pazopanib was similar between the two groups.

**Conclusion.** PRINCIPAL confirms the efficacy and safety of pazopanib in patients with advanced/metastatic RCC in a real-world clinical setting. **The Oncologist** 2019;24:491–497

Implications for Practice: PRINCIPAL is the largest (*n* = 657) prospective, observational study of pazopanib in patients with advanced/metastatic renal cell carcinoma, to the authors' knowledge. Consistent with clinical trial results that often contain specific patient types, the PRINCIPAL study demonstrated that the effectiveness and safety of pazopanib is similarly safe and effective in patients with advanced kidney cancer in a real-world clinical setting. The PRINCIPAL study showed that patients with advanced kidney cancer who are treated with first-line pazopanib generally do not show disease progression for approximately 10 months and generally survive for nearly 30 months.

# Introduction \_\_

The pazopanib approval for the treatment of patients with advanced RCC was granted following the randomized phase III VEG105192 trial, in which pazopanib significantly improved

progression-free survival (PFS) versus placebo (median, 9.2 vs. 4.2 months; p < .0001) [1]. The subsequent randomized, phase IIIb, noninferiority COMPARZ trial demonstrated

Correspondence: Manuela Schmidinger, M.D., Medical University of Vienna, Waehringer Guertel 18-20, 1090 Wien, Vienna, Austria. Telephone: 43-140-400-44291; e-mail: manuela.schmidinger@meduniwien.ac.at Received November 15, 2018; accepted for publication February 4, 2019; published Online First on March 13, 2019. http://dx.doi.org/10.1634/theoncologist.2018-0787

comparable efficacy of first-line pazopanib and sunitinib in patients with advanced RCC, with health-related quality-of-life (HRQoL) analyses favoring pazopanib over sunitinib [2]. Since these clinical trials, real-world pazopanib studies in RCC are accumulating and providing information about patients under-represented or excluded from clinical trials (e.g., patients with poor performance status [PS] or nonmeasurable lesions) [3]. The PRINCIPAL study evaluated the efficacy and safety of pazopanib in a multinational, real-world clinical setting.

#### Subjects, Materials, and Methods

# **Study Design and Patients**

PRINCIPAL was a global, prospective, observational study of patients with advanced/metastatic RCC treated with front-line pazopanib. All patients provided informed consent, and the study was conducted in accordance with International Conference on Harmonisation Good Clinical Practice, patient privacy requirements, and ethical principles outlined in the Declaration of Helsinki 2008.

The study was designed to enroll ~500–700 patients over approximately 30 months. Sample size was chosen based on the expected precision for the outcomes of interest (<5% for PFS, overall survival [OS], and objective response rate [ORR]), and the feasibility of enrolling the desired patient population over the enrollment period. Consecutive patients meeting eligibility were enrolled and followed for 30 months or until premature discontinuation due to death, consent withdrawal, loss to follow-up, or study termination. Patients who permanently discontinued study treatment were followed for progression, survival, HRQoL, and efficacy of subsequent therapies for up to 30 months. Patients completed the study if 30 months of follow-up was conducted, or if the patient died during study treatment or the follow-up period.

Patients aged ≥18 years at enrollment with a diagnosis of advanced or metastatic RCC of clear cell or predominant clear cell histology and a clinical decision to initiate treatment with pazopanib within 30 days of enrollment were eligible for the study. No prior exposure to multikinase inhibitor or anti-vascular endothelial growth factor (VEGF) inhibitor for advanced or metastatic disease was allowed.

Patients were assessed to determine whether they met key eligibility criteria from the COMPARZ [2] study (clinical trial eligible [CTE]): RCC diagnosis with clear-cell component histology (predominantly clear cell); no prior systemic therapy (interferon- $\alpha$ , interleukin-2) for advanced or metastatic RCC; locally advanced (disease not amenable to curative surgery or radiation therapy) or metastatic RCC (equivalent to stage IV RCC according to American Joint Committee on Cancer staging); measurable disease per RECIST v1.1; Karnofsky Performance Score (KPS)  $\geq$ 70; age  $\geq$ 18 years; total serum calcium concentration < 12.0 mg/dL; adequate organ function at baseline (absolute neutrophil count  $\geq$ 1.5  $\times$  10 $^9$ /L, hemoglobin  $\geq$ 9 g/dL, platelets  $\geq$ 100  $\times$  10 $^9$ /L, total bilirubin  $\leq$ 1.5 $\times$  upper limit of normal [ULN], aspartate aminotransferase [AST] and alanine aminotransferase [ALT]  $\leq$ 2.5 $\times$  ULN);

no prior use of an investigational or licensed drug that targets VEGF or VEGF receptor (VEGFR); no history or clinical evidence of central nervous system metastases; and no coronary artery disease or cerebral artery disease at baseline. Patients in the PRINCIPAL study who did not meet COMPARZ eligibility criteria or had missing data for any of the CTE criteria were considered the non-clinical-trial-eligible (NCTE) population.

#### **Assessments**

There were no protocol-mandated visits or procedures. Patient demographics, medical history, and disease characteristics were obtained at the baseline visit, and follow-up information was collected approximately every 3 months ( $\pm 4$  weeks). Tumor responses were assessed according to local processes and physician's clinical judgement.

Primary efficacy measures were PFS, OS, and ORR defined as complete response (CR) or partial response (PR). Other primary objectives were to determine relative dose intensity (RDI) and its effect on treatment outcomes; change in HRQoL relative to baseline; and frequency of serious adverse events (SAEs) and adverse events of special interest (AESIs). AESIs were defined as any adverse event (AE) resulting in a dose modification or discontinuation, or any reports of new onset/worsened hypertension, cardiac or thyroid dysfunction, or evidence of liver toxicity. The final primary objective was to compare the population demographics, disease characteristics, and RCC treatment history of patients in the CTE population with those of patients in the NCTE population. Secondary study objectives were to evaluate efficacy, safety, and RDI in patient subgroups.

Patients who received ≥1 dose of pazopanib were evaluable for efficacy (PFS and OS) and safety analyses (All-Treated [AT] population). The Measurable Disease (MD) population comprised patients with measurable disease at baseline and was used for the analysis of ORR.

## Statistical Analysis

Continuous variables were reported as medians and ranges, and categorical variables were reported as number and percentage of the total population. Evaluations were based on point estimates and 95% confidence intervals (CIs), and efficacy and safety analyses were stratified by baseline patient characteristics such as Eastern Cooperative Oncology Group (ECOG) PS and histologic subtype, as appropriate. No formal hypothesis or statistical significance testing was planned. A multivariate logistic regression analysis was performed to investigate the impact of baseline demographics and disease characteristics on pazopanib exposure (RDI <85% vs. ≥85%). RDI was calculated as the ratio of average daily dose of pazopanib to the recommended daily dose of pazopanib (expressed as a percentage). The mean RDI in the COMPARZ study was 84%; thus, the cutoff of 85% was used for the PRINCIPAL analysis. To explore the impact of baseline demographics, disease characteristics, and RDI on efficacy outcomes, a logistic regression model (for overall response) and Cox proportional hazards regression models (for PFS and OS) were used.



#### RESULTS

#### **Patients**

Of the 657 patients in the AT population, 501 (76.3%) completed the study, 62 (9.4%) were lost to follow up, and 55 (8.4%) withdrew consent (supplemental online Table 1). The median age was 66 years (range, 22–90), and approximately two thirds of patients were male (supplemental online Table 2). The percentage of all patients with favorable/intermediate/poor risk was 3.8%/55.3%/13.9%, respectively, per Memorial Sloan Kettering Cancer Center (MSKCC) criteria and 5.0%/52.2%/23.3% per International Metastatic Renal Cell Database Consortium (IMDC) criteria. MSKCC and IMDC risk data were missing for 27.1% and 19.5% of patients, respectively. Follow-up assessment data were available for 74% of patients at 3 months, 43% at 12 months, 27% at 24 months, and 20% at ≥30 months (supplemental online Table 3).

#### **Efficacy**

In the AT population, median PFS was 10.3 months (95% CI, 9.2–12.0) and median OS was 29.9 months (24.7 to not reached [NR]; supplemental online Fig. 1).

Event-free probability estimates for PFS at 6, 12, and 24 months were 0.673 (95% CI, 0.631–0.711), 0.453 (95% CI, 0.409–0.496), and 0.269 (95% CI, 0.229–0.310), respectively. Event-free probability estimates for OS at 6, 12, and 24 months were 0.846 (95% CI, 0.815–0.872), 0.722 (95% CI, 0.684–0.756), and 0.552 (95% CI, 0.509–0.592), respectively. The ORR was 30.3%, and the median duration of response and median time to response were 11 months (95% CI, 8.6–14.6) and 3 months (95% CI, 2.9–3.1), respectively (MD population). Subgroup efficacy analyses (PFS, OS, and ORR) suggested inferior outcomes in patients with ECOG PS  $\geq$ 2, poor MSKCC or IMDC risk grouping, no prior nephrectomy, and cytokine pretreatment (Table 1).

Cox regression analyses supported significantly shorter PFS and OS in patients with ECOG PS  $\geq 2$  (vs. <2; p < .001 and p < .001, respectively) and poor MSKCC risk (vs. favorable risk; p = .003 and p = .001), and significantly longer PFS and OS in patients who received prior nephrectomy (vs. no prior nephrectomy; p = .034 and p < .001; supplemental online Table 4). Intermediate MSKCC risk grouping (vs. favorable risk; p = .039) was associated with significantly shorter OS, but this difference was not statistically significant for PFS. Similarly, the odds of overall response (CR/PR) were significantly lower in patients with poor MSKCC risk (vs. favorable risk; p = .020); furthermore, treatmentnaive patients had significantly higher odds of overall response (vs. cytokine pretreated; p = .042).

#### Safety

At least one AE was reported by 486 (74%) patients overall, and the most commonly reported AEs were hypertension (22.8%), diarrhea (12.8%), increased ALT (11.1%), increased AST (7.0%), hypothyroidism (6.1%), increased blood thyroid-stimulating hormone (6.4%), and nausea (5.6%). Treatment-related SAEs were reported by 64 (9.7%) patients. Overall, 1,079 nonserious AESIs occurred in 399 (60.7%) patients,

including new-onset or worsened hypertension (n = 160), evidence of liver toxicity (n = 137), thyroid dysfunction (n = 90), cardiac dysfunction (n = 16), and any other event resulting in pazopanib dose modification or discontinuation (n = 233; Table 2).

#### Pazopanib Exposure

Most patients (84.0%) started on the standard daily pazopanib dose of 800 mg and were treated with a median average total daily dose of 800 mg over the study period (Table 3). Patients who received lower than the 800 mg daily dose of pazopanib had longer durations of pazopanib exposure. The median duration of exposure (without dose interruption) was 6.9 months, 11.1 months, 9.6 months, and 12.7 months for pazopanib doses of 800 mg, 600 mg, 400 mg, and 200 mg, respectively (supplemental online Table 5).

The percentage of patients with an RDI <85% was 41.4% overall. The multivariate logistic regression analysis identified poor MSKCC risk classification (vs. favorable risk; p = .046) as a baseline patient characteristic associated with significantly greater odds of receiving  $\geq$ 85% (vs. <85%) RDI. The percentage of patients with favorable/intermediate/poor MSKCC risk who received an RDI <85% was 52%/43.8%/26.4%, respectively. Conversely, baseline characteristics associated with significantly reduced odds of receiving an RDI  $\geq$ 85% included treatment-naive status (vs. cytokine pretreatment; p = .038) and coronary artery disease (vs. no coronary artery disease; p = .001; supplemental online Table 6).

Approximately half of the patients underwent dose/regimen change or interruptions (Table 3), and AEs were the primary reason for dose change or interruption (93.8%). Pazopanib was discontinued in 78.1% of patients overall, owing primarily to disease progression (44.3%) and AEs (14.5%; Table 3). The most common AE leading to treatment discontinuation was hepatotoxicity (n = 7; 1.1%).

# **Comparison of CTE and NCTE Populations**

Ninety-seven (14.8%) and 560 (85.2%) patients were included in the CTE and NCTE populations, respectively. The primary reasons for ineligibility to enter a clinical trial were the absence of data on measurable disease per RECIST v1.1 (n = 177; 100 patients had extant measurable lesion that was not consistent with RECIST v1.1 and 77 patients had no existence of measurable lesion), presence of coronary artery disease at baseline visit (n = 51), prior systemic therapy for advanced/metastatic RCC (n = 38), and history or clinical evidence of central nervous system metastases (n = 31; supplemental online Table 7).

For patients in the CTE and NCTE populations, the percentage with favorable/intermediate/poor risk was 7.2%/75.3%/11.3% and 3.2%/51.8%/14.3%, respectively, per MSKCC criteria and 7.2%/70.1%/22.7% and 4.6%/49.1%/23.4%, respectively, per IMDC criteria (supplemental online Table 1). However, data for MSKCC and IMDC risk classification were missing for 172 (30.7%) and 128 (22.9%) patients in the NCTE population, respectively. Other baseline disease characteristics such as number and location of metastatic sites and prior treatments were generally comparable between the CTE and NCTE populations. For the CTE and NCTE

Table 1. Efficacy (PFS, OS, and ORR) in patient subgroups

	PFS <sup>a</sup>		OS <sup>a</sup>		
Subgroup	Events, n/N	Median (95% CI), months	Events, n/N	Median (95% CI), months	– ORR <sup>b</sup> , <i>n/N</i> (%)
Overall	386/655	10.3 (9.2–12.0)	284/657	29.9 (24.7-NR)	168/554 (30.3)
Age at enrollment					
<65 years	159/279	11.6 (9.2–14.8)	111/281	33.9 (27.9–33.9)	77/237 (32.5)
≥65 years	227/376	9.9 (8.4–11.1)	173/376	25.9 (20.2-NR)	91/317 (28.7)
Sex					
Male	269/447	11.1 (9.5–12.6)	195/449	29.9 (24.3-NR)	113/381 (29.7)
Female	117/208	9.3 (7.0-11.0)	89/208	28.2 (20.3-NR)	55/173 (31.8)
Baseline ECOG PS					
<2	310/541	11.2 (9.8–13.2)	217/542	33.9 (28.6-NR)	147/455 (32.3)
≥2	40/50	2.8 (2.2-3.2)	38/51	5.5 (2.4–10.4)	6/45 (13.3)
Missing	36/64	11.4 (5.9-13.8)	29/64	25.9 (16.4-32.9)	15/54 (27.8)
Baseline MSKCC risk					
Favorable	10/24	25.4 (12.0-NR)	4/25	NR (NR-NR)	10/21 (47.6)
Intermediate	210/361	11.2 (9.5–13.7)	147/361	33.9 (26.9-NR)	102/313 (32.6)
Poor	71/91	4.2 (2.9-6.7)	65/91	9.6 (5.9-14.6)	14/73 (19.2)
Missing	95/179	12.1 (9.8–16.1)	68/180	32.9 (24.7-NR)	42/147 (28.6)
Baseline IMDC risk					
Favorable	18/32	25.4 (12.0-30.8)	6/33	NR (NR-NR)	14/27 (51.9)
Intermediate	196/341	11.6 (9.8–14.1)	127/341	32.9 (30.5-NR)	96/295 (32.5)
Poor	109/152	5.9 (4.1-8.0)	100/153	12.3 (8.7–16.4)	25/122 (20.5)
Missing	63/130	13.3 (9.2-17.4)	51/130	28.2 (22.3-NR)	33/110 (30.0)
Prior nephrectomy					
Yes	294/509	11.5 (10.2–13.8)	197/511	32.9 (29.7-NR)	141/427 (33.0)
No	88/141	6.7 (5.4–9.2)	84/141	14.8 (11.3-20.2)	23/122 (18.9)
Unknown	4/5	9.1 (2.2-14.1)	3/5	20.5 (7.6-NR)	4/5 (80.0)
Prior treatment					
Cytokine	32/37	6.7 (3.7–12.0)	20/38	22.0 (7.8-NR)	3/32 (9.4)
Treatment-naive	354/618	10.7 (9.5–12.5)	264/619	30.5 (24.7-NR)	165/522 (31.6)
RDI					
<85%	147/271	13.8 (11.0–17.3)	98/272	33.9 (30.5-NR)	77/225 (34.2)
≥85%	239/384	8.6 (7.0–10.3)	186/385	23.7 (18.5–29.6)	91/326 (27.9)
Eligibility for participation in clinical trial					
Yes (CTE population)	60/96	9.6 (7.8–13.2)	47/97	26.3 (16.9-NR)	32/97 (33.0)
No (NCTE population)	326/559	10.7 (9.2-12.3)	237/560	32.9 (24.5-NR)	136/457 (29.8)

<sup>&</sup>lt;sup>a</sup>Analysis conducted in All Treated (AT) population.

Abbreviations: CTE, clinical trial eligible; ECOG PS, Eastern Cooperative Oncology Group performance status; IMDC, International Metastatic Renal Cell Database Consortium; MSKCC, Memorial Sloan Kettering Cancer Center; NCTE, non-clinical trial eligible; NR, not reached; ORR, overall response rate; OS, overall survival; PFS, progression-free survival; RDI, relative dose intensity.

populations, the median average daily dose was reported as 800 mg and 733 mg, respectively, and the percentage of patients with an RDI <85% was 29.9% and 43.4%, respectively (Table 3). Efficacy was similar between the CTE and NCTE populations (Table 1). The frequency of all-grade AEs was 64.9% and 75.5%, and the frequency of grade  $\geq$  3 AEs was 32.0% and 44.5%, in the respective CTE and NCTE populations.

## **Efficacy in the NCTE Population**

In the subgroup of patients within the NCTE population with no existence of a measurable lesion (n = 77; 13.8%), median PFS was 10.7 (95% CI, 6.4–18.0); median OS was not evaluable (supplemental online Table 8). The ORR was 20.8%, and the median duration of response and median time to response were 7 months (95% CI, 4.2–11.8) and 3 months (95% CI, 2.6–4.3), respectively (MD population).



<sup>&</sup>lt;sup>b</sup>Analysis conducted in Measurable Disease (MD) population.

Table 2. Nonserious AESIs

AESIs	Total no. of events	Patients with AESI <sup>a</sup> , n (%)
Any AESI	1,079	399 (60.7)
Evidence of liver toxicity	249	137 (20.9)
New onset or worsened hypertension	199	160 (24.4)
Thyroid dysfunction	108	90 (13.7)
Cardiac dysfunction	18	16 (2.4)
Any other event resulting in pazopanib dose modification or discontinuation	505	233 (35.5)

<sup>&</sup>lt;sup>a</sup>All treated patients (n = 657).

Abbreviations: AESIs, adverse events of special interest.

Among treatment-naive patients (n = 522; 93.2%), median PFS was 10.8 (95% CI, 9.5–12.8) and median OS was 32.9 (95% CI, 24.7–NR) The ORR was 31.3%, and the median duration of response and median time to response were 9 months

(95% CI, 7.1–13.8) and 3 months (95% CI, 2.9–3.1). Subgroup efficacy analyses (PFS, OS, and ORR) based on concomitant disease at baseline did not identify any associations between the presence/absence of comorbidity at baseline and efficacy outcomes within the NCTE population.

#### **Subsequent Treatments**

The most common subsequent therapeutic procedure since study start was radiotherapy (14.8%; supplemental online Table 9). Following pazopanib discontinuation, 45.1% of patients received subsequent systemic therapy for RCC, the most common being sunitinib (25.6%), everolimus (25.1%), and axitinib (21.6%; supplemental online Table 9).

#### DISCUSSION

This large, prospective, observational study confirmed the efficacy and favorable safety profile of pazopanib in patients with advanced/metastatic RCC. The median PFS (10.3 months) and median OS (29.9 months) in this real-

Table 3. Pazopanib exposure

Parameter	CTE (n = 97)	NCTE (n = 560)	All patients (n = 657)
Starting daily dose, n (%)			
200 mg	0	3 (0.5)	3 (0.5)
400 mg	5 (5.2)	58 (10.4)	63 (9.6)
600 mg	5 (5.2)	34 (6.1)	39 (5.9)
800 mg	87 (89.7)	465 (83.0)	552 (84.0)
Once-daily frequency of total starting daily dose, n (%)	97 (100.0)	551 (98.4)	648 (98.6)
Average total daily dose, mg, median (range)	800 (250–800)	733 (200–1,067)	800 (200–1,067)
RDI %, median (range)	100 (31–100)	91.7 (25–133)	100 (25–133)
RDI <85%, n (%)	29 (29.9)	243 (43.4)	272 (41.4)
Patients with any dose/regimen changes or interruption, $n$ (%)	43 (44.3)	298 (53.2)	341 (51.9)
Number of dose/regimen change(s) or interruptions, n (%)			
0	54 (55.7)	262 (46.8)	316 (48.1)
1	17 (17.5)	131 (23.4)	148 (22.5)
2	8 (8.2)	81 (14.5)	89 (13.5)
≥3	18 (18.6)	86 (15.4)	104 (15.8)
Primary reason for dose change/interruption, n (%)			
AE	81 (83.5)	535 (95.5)	616 (93.8)
Toxicity resolved	8 (8.2)	46 (8.2)	54 (8.2)
Other	24 (24.7)	225 (40.2)	249 (37.9)
Patients with dose/regimen discontinuation, n (%)	68 (70.1)	445 (79.5)	513 (78.1)
Primary reason for dose discontinuation, n (%)			
Death	6 (6.2)	43 (7.7)	49 (7.5)
Disease progression	41 (42.3)	250 (44.6)	291 (44.3)
Patient decision	3 (3.1)	23 (4.1)	26 (4.0)
Lost to follow-up	2 (2.1)	12 (2.1)	14 (2.1)
AE	8 (8.2)	87 (15.5)	95 (14.5)
Presurgical procedure	0	5 (0.9)	5 (0.8)
Other <sup>a</sup>	9 (9.3)	35 (6.3)	44 (6.7)

<sup>&</sup>lt;sup>a</sup>This may include patients who discontinued due to reaching the end of the 30-month follow-up period. Abbreviations: AE, adverse event; CTE, clinical trial eligible; NCTE, non-clinical trial eligible; RDI, relative dose intensity.

world study were comparable to clinical trial data of pazopanib in advanced/metastatic RCC [1, 2, 4, 5]. Median OS within each MSKCC or IMDC risk group was longer than expected based on previous clinical trial and real-world data of pazopanib [5, 6]. One potential explanation for the prolonged survival within each risk group could be the benefit gained from subsequent treatment with newer agents (e.g., axitinib, nivolumab, and cabozantinib) [7, 8]. However, fewer than 10% of patients received subsequent treatment with agents other than the VEGF- or mammalian target of rapamycin-targeted therapies that have all been available since 2012 or earlier, suggesting that improved prolonged survival is not due to agents like nivolumab and cabozantinib.

Patients in this real-world study were separated into two groups based on eligibility criteria from the COMPARZ trial. Although there appeared to be no stark differences in PFS or OS between the CTE and NCTE populations, outcomes were numerically better in the NCTE population. This is contrary to previous real-world analyses of patients with advanced RCC who generally have a less favorable MSKCC risk profile and worse PS compared with clinical trial participants [3], as well as significantly poorer survival outcomes (PFS and OS) with first-line systemic or VEGF-targeted therapies [9] in trial-ineligible patients.

Time from diagnosis of locally advanced/metastatic RCC to treatment was <12 months in the overall patient population; in the NCTE population, it was double that in the CTE population. Time to treatment was similar between the NCTE population and overall population because of the larger sample size of NCTE versus CTE relative to the overall population. Importantly, patients in the CTE population, by definition, were not cytokine pretreated, which may have contributed to the imbalance in time to treatment between the CTE and NCTE populations. In the NCTE population, cytokine-pretreated patients had poor outcomes (PFS, OS, and ORR) compared with treatment-naive patients; however, as these patients made up a small proportion of the NCTE population (5.2%), it is difficult to make any meaningful conclusions about the effect of prior treatment on efficacy outcomes.

Although NCTE patients appeared to have a slightly poorer risk profile compared with CTE patients, data for risk group classifications were unavailable for a large percentage of patients in the NCTE population (30.7% for MSKCC and 22.9% for IMDC), limiting the comparisons and interpretation that can be made between trial-eligible and trial-ineligible patients in PRINCIPAL versus other trials. Moreover, although Cox regression analyses were not performed for the NCTE population, baseline comorbidity did not appear to be associated with worse survival outcomes (PFS and OS) outcomes for this group of patients, suggesting that, from a safety and efficacy perspective, patients with comorbidities who are normally excluded from clinical trials can be treated with pazopanib and achieve comparable outcomes as clinical-trial-eligible patients.

Notably, a major reason for NCTE in the current study was absence of data on measurable disease per RECIST v1.1 (almost one third of patients; 100 patients had extant measurable lesion that was not consistent with RECIST v1.1, and 77 patients had no existence of measurable lesion), whereas

other real-world studies did not include measurable disease as a criterion for the trial eligible population [9, 10]. These methodological differences may have contributed to the discrepancy in results between studies.

Baseline patient characteristics associated with a significant PFS and OS benefit included ECOG PS <2 (vs. ≥2), favorable MSKCC risk (vs. poor MSKCC risk), and prior nephrectomy (vs. no prior nephrectomy). PS (KPS or ECOG) and prior nephrectomy are well-established prognostic factors in RCC and have been used to stratify patients during treatment randomization in trials such as COMPARZ [2]. The MSKCC prognostic risk grouping, which comprises five pretreatment clinical factors to split patients into three risk groups (favorable, intermediate, or poor), was first validated to predict survival with first-line interferon- $\alpha$  in advanced RCC [11]. Since that time, MSKCC risk grouping has also been validated for use with first-line VEGFR-tyrosine kinase inhibitor therapy [12]. Poor MSKCC risk (vs. favorable risk) grouping was also associated with a significantly greater chance of receiving an RDI ≥85% of pazopanib, suggesting that although favorable-risk patients were more likely to receive a reduced dose, efficacy was not compromised, and in fact appeared improved. It may be that patients who required dose reduction were those who experienced a pharmacodynamic effect of therapy, whereas those capable of staying on full dose did not. In support of this, a recent retrospective analysis of patients with metastatic RCC who experienced cumulative toxicity with sunitinib or pazopanib had significantly longer OS and PFS compared with patients experiencing one or no AEs [13]. It is also plausible that patients with a favorable PFS, and therefore longer pazopanib exposure, may have been at higher risk of dose reduction. This hypothesis is supported by data from a recent retrospective analysis of patients with pazopanib-treated softtissue sarcoma, which showed that patients with RDI <80% had a longer PFS than those with RDI >80% [14]. However, as the RDI analysis described herein focused on the effect of baseline characteristics on pazopanib dose intensity, it is not possible to speculate further on the effect of pazopanib exposure on RDI.

Pazopanib treatment was generally well tolerated. A relatively low frequency of hypertension (22.8%), diarrhea (12.8%), ALT elevations (11.1%), and AST elevations (7.0%) were observed compared with past clinical trial data [3, 15] This, in association with the HRQoL results (supplemental online Fig. 2), confirm good tolerance of pazopanib in an unselected, multinational population.

Limitations of this study include its observational nature and associated potential under-reporting of AEs, and no requirement to collect AEs other than AESIs and SAEs. Another limitation is the relatively low proportion of patients with follow-up tumor assessment data at ≥30 months (20%) as well as time to response, which may have been influenced by the recommended visit schedule of every 3 months.

#### Conclusion

The PRINCIPAL study is the largest prospective, observational study of pazopanib in patients with advanced/metastatic RCC to our knowledge, and the study results are consistent with



prior clinical data of pazopanib and confirm the efficacy and favorable safety profile of pazopanib treatment for this patient population in the real-world clinical setting.

#### ACKNOWLEDGMENTS

Support for third-party writing assistance for this manuscript, supplied by Chris Ontiveros, Ph.D. (ApotheCom, New York, NY), Julia Burke, Ph.D. (ApotheCom, Auckland, New Zealand), and Anthony Zucker, Ph.D. (ApotheCom, London, U.K.), was provided by Novartis. This work was supported by Novartis Pharmaceuticals, Inc. Novartis Pharmaceuticals, Inc. sponsored the study, collated the data, and conducted statistical analyses. Please contact Chaitali Babanrao Pisal at chaitali\_babanra. pisal@novartis.com for data availability.

#### **AUTHOR CONTRIBUTIONS**

Conception/design: Manuela Schmidinger, Aristotelis Bamias, Robert Hawkins, Petri Bono, Qasim Ahmad, Eric Jonasch

Collection and assembly of data: Manuela Schmidinger, Aristotelis Bamias, Giuseppe Procopio, Sergio Vázquez, Narayanan Srihari, Haralabos Kalofonos, Petri Bono, Qasim Ahmad

Data analysis and interpretation: Manuela Schmidinger, Aristotelis Bamias, Giuseppe Procopio, Robert Hawkins, Angel Rodriguez Sanchez, Narayanan Srihari, Petri Bono, Chaitali Babanrao Pisal, Yulia Hirschberg, Luca Dezzani, Qasim Ahmad, Eric Jonasch

Manuscript writing: Manuela Schmidinger, Aristotelis Bamias, Giuseppe Procopio, Robert Hawkins, Angel Rodriguez Sanchez, Sergio Vázquez, Narayanan Srihari, Haralabos Kalofonos, Petri Bono, Chaitali Babanrao Pisal, Yulia Hirschberg, Luca Dezzani, Qasim Ahmad, Eric Jonasch Final approval of manuscript: Manuela Schmidinger, Aristotelis Bamias, Giuseppe Procopio, Robert Hawkins, Angel Rodriguez Sanchez, Sergio Vázquez, Narayanan Srihari, Haralabos Kalofonos, Petri Bono, Chaitali Babanrao Pisal, Yulia Hirschberg, Luca Dezzani, Qasim Ahmad, Eric Jonasch

#### **DISCLOSURES**

Manuela Schmidinger: Bristol-Myers Squibb, Eisai, Exelixis, Ipsen, Novartis, Pfizer, Roche (C/A), Roche (RF), Pfizer, Roche (travel, accommodation, expenses); Aristotelis Bamias: Novartis, Bristol-Myers Squibb, Ipsen, Roche (SAB), Novartis, Roche, Bristol-Myers Squibb (RF), Novartis, Bristol-Myers Squibb (H); Giuseppe Procopio: Bristol-Myers Squibb, Ipsen, Merck Sharp & Dohme, Novartis, Pfizer (SAB); Robert Hawkins: Cellular Therapeutics, Immetacyte Limited (E, OI), Bristol-Myers Squibb, EUSA Pharma, GlaxoSmithKline, Ipsen, Novartis, Pfizer (H), Medical Research Council - Phage Antibody Patents (IP); Sergio Vázquez: Novartis, Pfizer, Bristol-Myers Squibb, Roche, Eusapharma, Astellas, Sanofi, Merck Sharp & Dohme, Janssen, Bayer, Eli Lilly and Company (SAB); Narayanan Srihari: Roche, Merck Sharp & Dohme (H); Haralabos Kalofonos: Amgen, Genesis, Janssen, Leo, Eli Lilly and Company, Merck, Merck Serono, Merck Sharp & Dohme, Novartis, Pfizer, Roche (C/A), Amgen, Bayer, Genesis, Janssen, Eli Lilly and Company, Merck Serono, Merck Sharp & Dohme, Novartis, Pfizer, Roche (RF), Enorasis, Novartis, Pfizer, Roche (travel, accommodation, expenses); Petri Bono: Novartis (RF), Pfizer, Novartis, Orion Pharma, Bristol-Myers Squibb, Merck Sharp & Dohme, Ipsen, Faron Pharmaceuticals (H): Yulia Hirschberg: Novartis (E); Luca Dezzani: Novartis (E); Qasim Ahmad: Novartis (E); Eric Jonasch: Exelixis, Novartis, Pfizer, Peloton (RF), ARMO, Eisai, Exelixis, Novartis, Pfizer, Peloton (C/A). The other authors indicated no financial relationships.

(C/A) Consulting/advisory relationship; (RF) Research funding; (E) Employment; (ET) Expert testimony; (H) Honoraria received; (OI) Ownership interests; (IP) Intellectual property rights/inventor/patent holder; (SAB) Scientific advisory board

#### References \_

- 1. Sternberg CN, Davis ID, Mardiak J et al. Pazopanib in locally advanced or metastatic renal cell carcinoma: Results of a randomized phase III trial. J Clin Oncol 2010;28:1061–1068.
- **2.** Motzer RJ, Hutson TE, Cella D et al. Pazopanib versus sunitinib in metastatic renal-cell carcinoma. N Engl J Med 2013;369:722–731.
- **3.** Mitchell AP, Harrison MR, Walker MS et al. Clinical trial participants with metastatic renal cell carcinoma differ from patients treated in realworld practice. J Oncol Pract 2015;11:491–497.
- **4.** Sternberg CN, Hawkins RE, Wagstaff J et al. A randomised, double-blind phase III study of pazopanib in patients with advanced and/or metastatic renal cell carcinoma: Final overall survival results and safety update. Eur J Cancer 2013;49:1287–1296.
- **5.** Motzer RJ, Hutson TE, McCann L et al. Overall survival in renal-cell carcinoma with pazopanib versus sunitinib. N Engl J Med 2014;370:1769–1770.
- **6.** Perez-Valderrama B, Arranz Arija JA, Rodriguez SA et al. Validation of the International Metastatic Renal-Cell Carcinoma Database Consortium (IMDC) prognostic model for first-line pazopanib in metastatic

renal carcinoma: The Spanish Oncologic Genitourinary Group (SOGUG) SPAZO study. Ann Oncol 2016; 27:706–711.

- 7. Motzer RJ, Escudier B, McDermott DF et al. Nivolumab versus everolimus in advanced renal-cell carcinoma. N Engl J Med 2015:373:1803–1813.
- **8.** Choueiri TK, Escudier B, Powles T et al. Cabozantinib versus everolimus in advanced renal cell carcinoma (METEOR): Final results from a randomised, open-label, phase 3 trial. Lancet Oncol 2016;17:917–927.
- **9.** Heng DY, Choueiri TK, Rini BI et al. Outcomes of patients with metastatic renal cell carcinoma that do not meet eligibility criteria for clinical trials. Ann Oncol 2014:25:149–154.
- **10.** Marschner N, Staehler M, Muller L et al. Survival of patients with advanced or metastatic renal cell carcinoma in routine practice differs from that in clinical trials-analyses from the German clinical RCC registry. Clin Genitourin Cancer 2017;15:e209–e215.
- **11.** Motzer RJ, Bacik J, Murphy BA et al. Interferon-alfa as a comparative treatment for clinical

trials of new therapies against advanced renal cell carcinoma. J Clin Oncol 2002;20:289–296.

- **12.** Motzer RJ, Escudier B, Bukowski R et al. Prognostic factors for survival in 1059 patients treated with sunitinib for metastatic renal cell carcinoma. Br J Cancer 2013;108:2470–2477.
- 13. lacovelli R, Verri E, Cossu Rocca M et al. Prognostic role of the cumulative toxicity in patients affected by metastatic renal cells carcinoma and treated with first-line tyrosine kinase inhibitors. Anticancer Drugs 2017;28: 206–212.
- 14. Nakano K, Funauchi Y, Hayakawa K et al. Relative dose intensity of induction-phase pazopanib treatment of soft tissue sarcoma: Its relationship with prognoses of pazopanib responders. J Clin Med 2019:8.
- **15.** Escudier B, Porta C, Bono P et al. Randomized, controlled, double-blind, cross-over trial assessing treatment preference for pazopanib versus sunitinib in patients with metastatic renal cell carcinoma: PISCES Study. J Clin Oncol 2014; 32:1412–1418.



See http://www.TheOncologist.com for supplemental material available online.